Open Access Full Text Article

ORIGINAL RESEARCH

Patient Preferences for Episodic Migraine Medications: A Discrete Choice Experiment of Self-Injectable Versus Oral Treatments Targeting Calcitonin Gene-Related Peptide Pathway

Chiara Whichello¹, Lars Viktrup², Oralee J Varnado², Matthew Quaife¹, Myrto Trapali¹, Antje Tockhorn-Heidenreich³

¹Evidera, London, UK; ²Eli Lilly and Company, Indianapolis, IN, USA; ³Eli Lilly and Company, Bracknell, UK

Correspondence: Antje Tockhorn-Heidenreich, Eli Lilly and Company Ltd, 8 Arlington Square West, Downshire Way, Bracknell, RG12 IPU 0044, UK, Tel +44 7901 92 76 75, Email tockhorn_antje@lilly.com

Purpose: To understand the trade-offs that patients with episodic migraine are willing to make between attributes of self-injectable calcitonin gene-related peptide (CGRP) monoclonal antibodies (mAbs) and oral small-molecule calcitonin gene-related peptide receptor antagonists (gepants).

Patients and Methods: This was an online discrete choice experiment survey among adults with episodic migraine (≥ 4 and <15 migraine headache days and <15 headache days per month over 3 months) in the United States. Hypothetical treatments were characterized by five attributes: chance of \geq 50% migraine reduction, impact on daily activities, onset of treatment effect, reduction in number of acute medications, and treatment administration. The attributes were selected based on insights gained from a previously conducted literature review and focus group study and aligned with the primary and secondary outcome measures in the double-blind Phase IV head-to-head clinical trial of a CGRP mAb versus gepant.

Results: 601 patients (mean age: 44.8 years) completed the survey. Treatment preferences differed significantly between patients. However, for all patients, the most important driver of treatment preferences was the chance of a \geq 50% reduction in monthly migraine headache days (relative attribute importance: 38.3%), followed by the impact on daily activities (23.5%), the onset of treatment effect (19.5%), the reduction in need for acute medication (15.4%), and finally the route of administration (3.4%). Patients were willing to consider a one-week delayed onset of treatment effect or one-day increased need for acute medication for a higher chance (by 2.06% and 2.65% respectively) of a \geq 50% reduction in monthly migraine headache days. Patients would trade a reduction of migraine's impact on daily activities from "extreme" to "moderate" or "minimal" with a lower chance (17.09%, 12.06% respectively) of halving the number of monthly migraine headache days.

Conclusion: A \geq 50% reduction in monthly migraine headache days was the most important treatment attribute for which participants were willing to trade against other attributes. The variation in treatment preferences between patients emphasizes the importance to align decision-making with individual patients' preferences.

Plain Language Summary: Patients seeking care for migraine have varying levels of migraine severity, disability, and different experience with prior treatment. In this study, we researched patients' preferences for different preventive treatments, by having 601 patients complete an online survey. The survey presented a 'discrete choice experiment' where hypothetical migraine treatments described by five treatment characteristics, including whether the treatment was an oral pill or an injection, and asked patients to choose between them. Our results suggested that the chance of reducing the number of migraine days each month by half was the most important therapeutic characteristic, for which patients were willing to exchange other treatment characteristics in order to have a greater chance of achieving this improvement. Patients also valued if treatments worked quickly or reduced the impact of migraine on everyday activities. Whether a treatment was given as an injection or oral pill did not affect preferences.

Keywords: gepants, self-injectable treatment, treatment preferences, preventive treatment, United States

839

© 2025 Whichello et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/ the work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraph 4.2 and 5 of our Terms (https://www.dovepress.com/terms.php).

Introduction

Migraine is a chronic debilitating neurologic disease, and the migraine impact according to migraine frequency, duration, and intensity may vary greatly among patients throughout the course of their illness.¹ Moreover, preventive therapies for migraine differ not only in terms of treatment efficacy and side effects but also with regards to onset of treatment effect, need for additional acute medication, functional improvement, cost, or route and frequency of administration.² Consequently, to optimize therapy at the individual level, patients with migraine and their healthcare professionals (HCP) need to examine and choose from a broad variety of treatment characteristics.^{2,3} In addition, previous studies have demonstrated heterogeneity in patients' treatment choices, suggesting that each patient places different values on distinct treatment features.^{3–11} Understanding choices among diverse treatments requires an understanding of what trade-offs patients are willing to make between treatment characteristics.

Traditional preventive therapies for migraine were initially approved for other indications, and though commonly used, their suboptimal efficacy, tolerability issues, and drug-drug interactions make their use challenging. ^{2,5,12} Since 2018, six new medications targeting calcitonin gene-related peptide (CGRP), a neuropeptide involved in migraine pathophysiology,¹³ or its receptor have been approved for migraine prevention by the United States (US) Food and Drug Administration (FDA). Moreover, consensus statements from the American Headache Society and the European Headache Foundation recommend considering CGRP-targeted medications as first-line therapy for the preventive treatment of migraine without a requirement for prior failure of other classes of migraine preventive treatment. ^{2,14,15} These comprise four monoclonal antibodies (mAbs) targeting CGRP (galcanezumab, fremanezumab, and eptinezumab) or its receptor (erenumab) and two small-molecule CGRP receptor antagonists (the "gepants" rimegepant and atogepant). Some contradicting evidence exists on the comparison between mAbs targeting CGRP and gepants. ^{16–18} A meta-analysis of 19 Phase III randomized-controlled clinical studies showed that all reduced mean monthly migraine days compared to placebo, with rimegepant reducing them by 0.8 days and galcanezumab by 2.3 days. ¹⁷ A single double-blind, double-dummy, head-to-head clinical trial of galcanezumab vs rimegepant in people with episodic migraine (EM) showed no statistical significant difference with >60% of patients experiencing \geq 50% reduction in monthly migraine headache days. However, 84% of the patients were preventive treatment naive.¹⁸

CGRP-targeted medications differ substantially in their dosing, frequency, and route of administration. Some CGRP mAbs are self-injectable subcutaneous formulations that are administered monthly (erenumab, fremanezumab, and galcanezumab)^{19–21} or once every three months (fremanezumab).²⁰ The CGRP mAb eptinezumab²² requires a healthcare provider to administer it as an intravenous infusion once every three months. The gepants atogepant²³ and rimegepant²⁴ are orally administered. Previous preference studies comparing mode of administration have reported that it is relatively less important than other treatment attributes; however, the relative importance of different dosing frequencies and routes of administration compared in these studies varied widely.^{3–5}

Some patient preference studies have been conducted following the advent of these newer treatments specifically developed for migraine.^{3–5,7,8} However, these studies did not directly compare attributes of self-injectable CGRP mAbs and oral gepants. Such comparison is needed to ensure that patients and HCPs are able to make optimal decisions that align with individual patient preferences. For example, patients may prefer one mode of administration over an alternative, but this may be compensated by better performance in other attributes, such as efficacy. The degree to which compensation is possible depends on the trade-offs that patients are willing to make. To address this evidence gap, we conducted this preference study to evaluate which treatment attributes of a self-injectable CGRP mAb or an oral gepant are most valued by patients deciding among different preventive treatments for EM and aimed to understand the trade-offs the patients are willing to make between these attributes.

Materials and Methods

Study Design

A web-based discrete choice experiment (DCE) was conducted among patients with migraine between April and October 2022 to quantify the relative importance they place on different treatment attributes of self-injectable CGRP mAbs and oral gepants and the trade-offs they are willing to make among those attributes. We also investigated preference

heterogeneity depending on patients' sociodemographic or clinical characteristics, as well as migraine diagnoses being selfreported or confirmed by a physician's diagnosis. The study design process began by defining treatment attributes and levels based on previous research evidence. The resulting DCE instrument was tested in qualitative interviews and quantitative pilot with patients before being completed by the study population.

Eligible for recruitment, were adults (\geq 18 years) living in the US with self-reported diagnosis of migraine by a physician or with a physician-provided confirmation of diagnosis (COD), if they had EM, as shown by their answers to a detailed clinical screener. EM was defined as \geq 4 and <15 migraine headache days and <15 headache days per month over 3 months (Supplemental Methods). Patients could self-report physicians' diagnosis for migraine, however, to balance competing concerns of recruitment feasibility, patient and HCP burden, and confidence that patients report their disease status accurately, a target recruitment quota of 50% was set so that half of the enrolled patients would submit a physician-provided confirmation of diagnosis (COD) in addition to self-reporting. This was an online or paper form confirming the diagnosis of episodic migraines and the number of migraine headache days and headache days the patient experiences per month. Eligible patients had at least moderate migraine severity, as evaluated by the Migraine Symptom Severity Score (MSSS; moderate severity was defined as: "scoring '3-5' on any of the first four 'pain' items and scoring 3-5 on the 'nausea' item or both the 'photophobia' and 'phonophobia' item. (Supplemental Methods)". A healthcare research organization specializing in internet-based panels recruited the study patients via online databases and panels, social media, and patient associations.

Ethics Approval and Informed Consent

The study was conducted in accordance with the Declaration of Helsinki. The study protocol was submitted to Ethical & Independent (E&I) Review Services, a fully accredited institutional review board (IRB), for central ethical approval. IRB approval for the initial submission was granted on 27 April 2021 (E&I study number: 21069–01), and IRB approval of translated documents (from English to Spanish and Puerto Rican Spanish) was granted on 14 May 2021 (E&I study number: 21069–01A). Following modifications to the survey (incorporating the qualitative pilot interview feedback), IRB approval was received for the amended protocol on 24 February 2022 (E&I study number: 21069–01B). Lastly, continuing review approval was granted on 8 June 2022 (E&I study number: 21069–02).

All patients reviewed and completed an online informed consent form before participating in the study. The DCE study was conducted according to best-practice guidelines concerning pharmacovigilance practices^{25,26} and preference-based analyses, as recommended by the International Society for Pharmacoeconomics and Outcomes Research.^{27–29}

DCE Survey

The first section of the survey introduced patients to the treatment attributes with examples and practice questions designed to engage patients with the DCE attributes and encourage attentive evaluation of the attributes. The second section of the survey contained 13 experimental and two non-experimental DCE choice tasks (Figure 1).

In the final survey section, patients were asked to complete sociodemographic and clinical questionnaires. Health literacy was also assessed with three questions from the Set of Brief Screening Questions (SBSQ),³⁰ and numeracy was assessed with five questions from the Numeracy Scale.^{31,32} Patients completed several patient-reported outcome (PRO) assessments to evaluate migraine-associated symptoms. The Migraine-Specific Quality of Life Questionnaire (MSQ) measures the impact of migraines on daily activities.^{33,34} The MSQ score assesses patient functioning on a scale from 0 to 100, with scores of <40 indicating extreme impairment and scores of 85–100 indicating no or minimal impairment.³⁵ The Patient Global Impression of Severity (PGI-S) measures severity of illness, with scores ranging from 1 (normal, not at all ill) to 7 (extremely ill).³⁶ The Migraine Disability Assessment questionnaire (MIDAS) measures headache-related disability on a scale ranging from little or no disability (0–5) to severe disability (>20).³⁷ The MSSS score ranges from 8 to 40 with higher scores indicating more severe disability.³⁸

DCE Choice Tasks

In each choice task, patients were asked to choose between hypothetical treatment alternatives that were characterized by a set of attributes with different levels (eg, 40%, 55%, or 70% chance of reduction in monthly migraine headache days) (Table 1). The order of experimental choice tasks was randomized across patients,³⁹ (Supplemental Statistics).

	Treatment A	Treatment B
How you take the treatment	Orally dissolving pill every other day	Self-injection once a month
Chance of reducing the number of migraine headache days per month that you experience by at least half	***** *******************************	*** *********************************
Time it takes for the medication to work	2 weeks	1 month
Impact of migraines on daily activities	Moderately impaired by migraine	Minimally impaired by migraine
Reduction in the number of migraine days you take acute medication (to treat migraine symptoms)	Days -6 6 fewer days	Days 4 fewer days
Choice		

Figure I Example of a DCE choice task^a.

Note: ^aPatients were presented with 13 pairs of such hypothetical scenarios and were asked to choose which one they would prefer.

Abbreviation: DCE, discrete choice experiment.

DCE attributes were selected based on insights gained from a previously conducted literature review and focus group study,⁸ which explored patients' preferences for attributes of preventive migraine treatments. Additionally, attributes were aligned with the primary and secondary outcome measures in the double-blind phase IV head-to-head clinical trial of galcanezumab versus rimegepant (Table 1). Attributes did not include safety or tolerability as CGRP mAbs and oral gepants are generally well tolerated and no available study data have suggested a major difference in safety between CGRP mAbs and oral gepants.¹⁶ Attribute levels were selected to cover the likely ranges of outcome measures reported in clinical trials of galcanezumab^{42–44} and rimegepant.⁴⁰ A D-efficient experimental design was used (Ngene[®] software version 1.2.1, ChoiceMetrics, Australia) to select the subset of choice tasks that could be used to estimate each attribute's effect on preferences.^{46–48}

An initial version of the DCE instrument was developed and tested in qualitative and quantitative pilot tests (Supplemental Methods/Results). Qualitative pilot interviews were conducted with 13 patients in web-based teleconferences and aimed to assess whether patients considered the selected DCE attributes and levels relevant, tradeable, and understandable. The complexity, clarity, length, wording, and completeness of the questions were also assessed. The attribute presentation and wording were then refined accordingly. Quantitative pilot testing followed in 101 patients and aimed to evaluate whether attribute levels covered the preference-relevant range, identify other meaningful parameter estimates, and update attribute levels where necessary.²⁹ Because no changes were made to the attributes, levels or experimental design at this stage, quantitative pilot testing data were added to the main DCE sample as recruitment proceeded.

Attributes	Levels	Rationale for selection of levels
How you take the treatment	Self-injection once per month Orally dissolving pill every other day Oral pill every day	Levels reflect the different modes of administration for galcanezumab (monthly self-injection) and the oral gepants rimegepant (orally dissolving tablet every other day) and atogepant (oral pill every day) ^{21,40,41}
Chance of reducing the number of migraine	40%	Levels reflect results from previous clinical trials measuring
headache days per month that you	55%	a \geq 50% reduction in monthly migraine headache days in
experience by at least half	70%	EVOLVE-1 (galcanezumab 120 mg: 62.3%, 240 mg: 60.9%; placebo: 38.6%), EVOLVE-2 (galcanezumab 120 mg: 59.3%, 240 mg: 56.5%; placebo: 36.0%), ^{42,43} and CONQUER ITT EM (rimegepant 120 mg: 41.8%; placebo 17.1%). ⁴⁴
Time it takes for the medication to work	2 weeks	Levels were informed by reduction in migraine headache days
	I month	within the first month of treatment and over time in
	2 months	a combined analysis of EVOLVE-1 and EVOLVE 2 trials. ⁴⁴ Both tested doses of galcanezumab (120 and 240 mg) achieved a statistically significant reduction in the number of monthly migraine headache days beginning at month 1 and continuing through month 6. A 240 mg loading dose of galcanezumab had a significant effect at week 1 that continued throughout the remaining weeks of month 1
Impact of migraines on daily activities	Minimally impaired by migraine	Levels correspond to changes from baseline in the role function-
	Moderately impaired by migraine Extremely impaired by migraine	restrictive subdomain of the MSQ. The role function-restrictive domain measures the impact of migraines on daily activities ^{33,34} and has previously been used in clinical trials of galcanezumab. ⁴⁵ The lower-, middle-, and higher-level categories (minimally, moderately, and extremely impaired) were chosen as attribute levels to ensure sufficient differentiation of levels across the spectrum of functioning and disability
Reduction in the number of days you take	2 fewer days	Levels reflect the reduction in monthly migraine headache
acute medication (to treat migraine	4 fewer days	days requiring acute medication use in EVOLVE-1
symptoms)	6 fewer days	(galcanezumab 120 mg: -4.0 days; 240 mg: -3.8 days; placebo: -2.2 days) ⁴³ and EVOLVE-2 (galcanezumab 120 mg: -3.7 days; 240 mg: -3.6 days; placebo: -1.9 days).Levels reflect the reduction in monthly migraine headache days requiring acute medication use in EVOLVE-1 (120 mg: -4.0 days; 240 mg: -3.8 days; placebo: -2.2 days) ⁴³ and EVOLVE-2 (120 mg: -3.7 days; 240 mg: -3.6 days; placebo: -1.9 days). ^{42,43}

Table I Treatment Attributes and Levels Used in the DCE

Abbreviations: DCE, discrete choice experiment; MSQ, Migraine-Specific Quality of Life Questionnaire.

Data Quality

Patients completed two non-experimental choice tasks, stability and the dominance tests, which were not included in the final DCE analysis but were used as data quality indicators.^{49,50} The stability test repeated the third choice task at the end of the DCE survey to check whether the respondent provided the same answer twice. This was followed by the dominance test, evaluating the rationality in the choice behavior of patients ie, whether they would choose the superior option among two choices: one with favorable levels for all attributes and an alternative with less favorable levels for chance of migraine reduction, impact on daily activities and reduction in acute medication.^{49,50} Finally, response time was measured to quantify the time patients took to complete all 13 experimental tasks.

Data Analysis

Descriptive statistics were used to summarize patient characteristics and data on health literacy, numeracy, dominance and stability tests, dominated decision-making, serial non-participation, and response time. DCE choice data were analyzed in a random utility maximization framework.^{51–53} Mixed multinomial logit (MXL) models were chosen to summarize the main analyses due to model fit and flexibility in assumptions.^{54,55} MXL models were estimated with fully correlated covariance matrices; this was done in a Bayesian framework to make computation feasible. Marginal utilities were calculated as a measure of attribute desirability, where a higher marginal utility for a specific attribute level indicates a greater likelihood that a treatment scenario including that attribute level will be chosen, everything else being equal. Relative attribute importance (RAI) scores quantify the maximum percentage contribution of each attribute level with the highest preference weight and the level with the lowest preference weight, relative to the preference weight of the other DCE attributes. RAI values for all attributes total 100%. Marginal rates of substitution were estimated to determine the willingness of a patient to exchange a proportion of one attribute for a proportion of another attribute that is equally satisfying (Supplemental Statistics).⁵⁵ Subgroup analyses, by physician's COD availability, demographics, and clinical characteristics, were conducted to examine whether patient characteristics rather than treatment attributes influenced preference heterogeneity.

Results

Out of the 5000 invited potentially eligible individuals, approximately 3000 agreed to participate in the study (Figure 2). Of these, 784 were eligible, and 601 completed the survey. The mean age of the patients was 44.8 years (standard deviation [SD]=13.0, range 19–85) (Table 2). Patients were mainly male (67.6%) and about 82.7% had obtained at least some college education. Three-hundred patients provided physician's COD, and 301 provided a self-reported diagnosis only. To account for potential scale heterogeneity across the COD and no-COD samples and allow pooling, MXL models were estimated with fully correlated covariance matrices.

About half of patients (51.0%) had experienced more than eight migraine headache days per month (Table 2). A greater number of (12 to 14) migraine headache days per month were more frequent among patients with physician's COD than among those without physician's COD (43.3% vs 27.6%; <u>Supplemental Table 1</u>). Some patients (30.0%) had never previously used a preventive medication for migraine (Table 2). About half the patients (53.2%) had experience using a self-injectable medication for a health condition. Over half of the patients (57.8%) were "not at all" or "a little" afraid of injecting themselves with a medicine; however, 25.5% were "moderately" afraid, and 16.8% were "very" or "extremely" afraid (Table 2). Lack of fear of self-injecting a medicine was more frequently reported by patients with physician's COD than by those without physician's COD (48.3% vs 24.3%; <u>Supplemental Table 1</u>).

Results from the four PRO instruments indicated that the patients experienced moderate to severe migraine-related symptoms and disability (Table 2); patients with a physician's COD scored significantly higher on average than patients without physician's COD on every PRO instrument (<u>Supplemental Table 1</u>). Most patients had high health literacy (overall: 84%; physician's COD: 90% vs no physician's COD: 79%) and health numeracy (overall: 97%; physician's COD: 95% vs no physician's COD: 100%) scores (<u>Supplemental Table 2</u> and <u>3</u>).

Most patients made consistent choices in the stability test (80.9%) and passed the dominance test (88.4%) (<u>Supplemental Table 4</u>). Moreover, nearly all patients (91.3%) did not show dominant preferences toward any single attribute, indicating that they made trade-offs among different attributes. About half of the patients (50.7%) took \geq 3 minutes to complete the DCE survey.

Preference Elicitation

As shown in Figure 3, at least one marginal utility was significant for all attributes except treatment administration, which did not significantly affect respondent choices. In Figure 4 we see that the chance of a \geq 50% reduction in monthly migraine days was the most important driver of patients' preferences for a migraine treatment (RAI: 38.3%; 95% confidence interval [CI]: 34.0–42.4), and treatment administration was the least important (RAI: 3.4%; 95% CI: 0.7–7.2)



Figure 2 Patient disposition flow chart.

Note: ^aPatients were required to have a diagnosis of EM, defined as ≥ 4 and <15 migraine headache days and <15 headache days per month over 3 months; ^bA target recruitment quota of 50% was set for patients with a physician-provided confirmation of an EM diagnosis;^c Patients with Alzheimer's disease/dementia, Parkinson's disease, or schizophrenia were excluded from the study.

Abbreviations: DCE, discrete choice experiment; EM, episodic migraine.

(Figure 4). In other words, having the chance to reduce the number of monthly migraine days by \geq 50% was 11.2 times more important to patients than the way the treatment was administered. Impact on daily activities was also highly valued by patients (RAI: 23.5%; 95% CI: 19.0–27.8). Onset of treatment effect (RAI: 19.5; 95% CI: 16.4–22.9) and reduction in acute medication (RAI: 15.4%; 95% CI: 12.1–18.7) were comparable in their importance to patients.

Treatment Attribute Trade-Offs

The average trade-offs that patients were willing to make between different aspects of migraine treatments are presented in Figure 5. Based on these, patients would trade a reduction of migraine's impact on daily activities from "extreme" to "minimal" for a 17.09% (95% CI: 12.93–21.25) decrease in the chance of halving the number of migraine headache days per month. Similarly, patients would trade a treatment that reduced the impact of migraines on daily activities from "extreme" to "moderate" for a 12.06% (95% CI: 9.37–14.75) decrease in the chance that the treatment would reduce the number of migraine headache days per month. Patients would trade a one-week delay in onset of treatment effect for a 2.06% (95% CI: 1.60–2.53) higher chance of reducing the number of migraine headache days by \geq 50%. Finally,

Characteristic	N=601	
Age (years)		
Mean (SD)	44.8 (13.0)	
Range (min, max)	19, 85	
Sex (male), n (%)		
Educational background, n (%ª)		
High school or less	104 (17.3)	
Some college / university	194 (32.3)	
College / university	185 (30.8)	
Post graduate degree	118 (19.6)	
Mean migraine headache days/month in past 3 months, n (%ª)		
4–7	294 (48.9)	
8–11	207 (34.4)	
12–14	100 (16.6)	
Mean headache days/month in the past 3 months, n (%ª)		
4–7	171 (28.5)	
8–11	217 (36.1)	
12–14	213 (35.4)	
Age at first diagnosis of migraine (years), mean (SD)		
Prior use of a self-injectable for any health condition, n (%)		
No	281 (46.8)	
Yes	320 (53.2)	
Fear of self-injection, n (% ^a)		
Not at all	218 (36.3)	
A little	129 (21.5)	
Moderately	153 (25.5)	
Very	67 (11.1)	
Extremely	34 (5.7)	
Prior use of preventive treatments for your migraine, n (% ^a)		
No, never used	180 (30.0)	
Yes, currently using ^b	272 (45.3)	
Yes, have used in the past ^b	173 (28.8)	
MSSS score, mean (SD)		
PGI-S score, mean (SD)		
MSQ score, mean (SD)		
MIDAS score, mean (SD)	41.6 (38.8	

Table 2 Patient Demographic and Clinical Characteristics

Notes: ^aPercentages calculated over 470 (the denominator) received responses; ^bthe two "Yes" categories were not mutually exclusive as both could be true for one patient.

Abbreviations: MIDAS, Migraine Disability Assessment; MSQ, Migraine-Specific Quality of Life Questionnaire; MSSS, Migraine Symptom Severity Score; PGI-S, Patient Global Impression of Severity; SD, standard deviation.

patients would trade a one-day increase in acute medication use for a 2.65% (95% CI: 1.96–3.35) increase in the chance of reducing the number of migraine headache days by \geq 50%.

Preference Variation by Patient Characteristics

Treatment administration had greater importance among subgroups defined by reluctance to self-inject, MIDAS migraine disability ranged from no disability to moderate disability, MSQ score 67–100, age \geq 65 years and 18–34 years, patient functioning, and than in the overall sample (<u>Supplemental Figure 1</u>). Therefore, patients with less headache-related disability or functional impairment placed greater relative importance on treatment administration than did patients in the overall sample (RAI for MIDAS little or no/mild/moderate scores: 20% vs 3.4%, p<0.01; RAI for MSQ score 67–100:



Figure 3 Marginal utilities derived from the mixed multinomial logit model (n=601).

Notes: The mean marginal utility and 95% CI of each attribute level, relative to the other levels included in the DCE, are shown. Estimates are marginal utilities that denote the effect of deviating from a reference level of on preferences. Positive mean estimates with a 95% CI >0 suggest an overall preference for the attribute level over the reference level. *** indicates p<0.001.

Abbreviations: Cl, confidence interval; DCE, discrete choice experiment; SE, standard error.

18% vs 3.4%; p<0.01). Additionally, patients aged 18 to 34 years and >65 years attributed more importance to treatment administration than did patients in the overall sample (RAI: 14% vs 3.4%, p<0.05; RAI: 17% vs 3.4%; p<0.05, respectively).

The importance of other treatment attributes varied among subgroups. Patients with "little or no", "mild", or "moderate" disability according to MIDAS scores attributed less importance to a treatment's impact on daily activities than did patients in the overall sample (RAI: 13% vs 23.5%; p<0.05). Furthermore, patients with physician's COD attributed more importance to a treatment's impact on daily activities compared to the overall sample (RAI: 30% vs 23.5%; p<0.01), whereas patients without physician's COD attributed less importance (RAI: 14% vs 23.5%; p<0.01). In addition, patients with "moderate", "very", or "extreme" reluctance to self-inject placed lower relative importance on a treatment's ability to reduce acute medication usage than did patients in the overall sample (RAI: 10% vs 15.4%; p<0.05).

Discussion

In this DCE study in patients with EM, the most important driver of patient treatment preferences was a \geq 50% reduction in monthly migraine headache days, followed by the impact on daily activities, the onset of treatment effect, the reduction in need for acute medication, and finally the route of administration. Patients were willing to trade off any other treatment attribute for a chance of reducing migraine headache days by \geq 50%, such as one-week delayed onset of treatment effect or a one-day increased need for acute medication. Patients would also consider a treatment with a lower chance of a \geq 50% reduction in monthly migraine days in exchange for a reduced impact of migraines on daily activities. A recent three-month, double-blind, phase IV head-to-head clinical trial (NCT05127486) compared galcanezumab and



Figure 4 Mean RAI scores.

Note: RAI scores were calculated to facilitate the interpretation of the marginal utility estimates. RAI scores are reported on a scale of 0 to 100%, where 100% reflects choices driven entirely by that attribute, and 0% reflects choices completely ignored for that attribute. The RAI score and 95% CI for each attribute are shown.

rimegepant for EM prevention.¹⁸ That trial used a \geq 50% reduction from baseline in monthly migraine headache days as the primary outcome, which the present study supports as a patient-relevant primary endpoint in clinical trials.

Mode of treatment administration (self-injectable or oral medication) was the least important attribute, of all others considered, to this DCE survey patients, who were unwilling to accept any loss of treatment efficacy in exchange for a different mode of treatment administration. This is consistent with a previous DCE survey of adults with a self-reported physician diagnosis of migraine in the US, which found that the mode and frequency of administration (ie, daily oral pill, monthly injections, or twice-monthly injections) had minimal effects on patient preferences for migraine treatments.⁵

Only 16.8% of respondents were "very" or "extremely" afraid of injecting themselves with a medicine, and respondents were broadly familiar with self-injectables as nearly half of patients (53.2%) said they were either currently using a self-injectable medication or had done so in the past, although this may not have necessarily been specific for migraine treatments. One possible explanation for this high prevalence of self-injectable use might be previous migraine treatment with CGRP mAb, given that most patients were treated with migraine preventive treatment before entering the study. Only 23.5% reported using erenumab, fremanezumab, galcanezumab, or eptinezumab currently or in the past. Therefore, previous migraine injectable experiences could have influenced the present responses. A marked variability in patient preferences for self-injectable or oral medication was reported in a previous survey of 601 adults with migraine, with approximately 50% of patients favoring self-injectable medication, 30% preferring oral medication, and 20% tending to choose either depending on other treatment attributes, such as dosing schedule or mechanism of needle removal.³ Another study found that patients exhibited an overall preference for an oral tablet every other day over a quarterly infusion, quarterly injection, or monthly injection; however, patient preferences were heterogeneous, in part because patients with previous CGRP mAb experience placed less importance on the mode of administration than other patients did.⁴ Adding to such previous findings, the present DCE study findings indicate that subgroups of patients who are inexperienced with self-injectable medications or fear self-injection may place a greater value on treatment administration than other patient subgroups do.



Figure 5 Willingness to trade attributes to reduce the number of headache days by half.

Note: Estimates of patients' willingness to exchange a proportion of one attribute for another equally satisfying were calculated using a mixed multinomial logit model that assumed log-normal distributions relative to the chance of reducing migraine headache days by half. Error bars represent 95% confidence intervals. *** indicates p-value <0.01%. Abbreviation: SE, standard error.

In this study, preferences differed significantly between patients, emphasizing the importance of decision-making that aligns with patients' individual treatment priorities. These findings are consistent with previous preference research in this field, highlighting the importance of HCPs considering patient preferences when recommending migraine treatments.⁷ Although the attributes were informed by previous literature and clinical endpoints, only five treatment characteristics were included. Therefore, the results and observations of this study are limited to these five attributes. This DCE study did not include safety and tolerability as treatment attributes; however, existing evidence does not suggest significant differences between self-injectable CGRP mAbs and oral gepants in these treatment characteristics. However, we included injection-related reactions in treatment attributes to ascertain the assessment of this unique side effect.

Selection bias is a potential limitation of this study, as the experiences and preferences of the sample may be systematically different from the general population of patients with migraine. For example, the male predominance among survey patients was surprising given the common epidemiological profile of patients with migraine, who are more often female.^{56,57} Patients were identified through a convenience sample drawn from an opt-in panel of individuals. To minimize selection bias, several recruitment methods were used, including recruitment through online panels, clinical organizations, or social media. Enrolling patients with self-reported physician diagnosis of migraine is a common practice in patients preference research.^{3–5,7,8} In this study, the overall sample was composed of patients who passed a detailed clinical screener, including the disease-specific MIDAS questionnaire, and were also asked to provide COD, and those who passed the screener but were not able to provide COD. Such diagnostic criteria are adequate for a web-based survey study assessing patient preferences. Furthermore, comprising the overall sample of these two groups of patients balanced competing concerns of recruitment feasibility, patient and physician burden, and confidence that patients report their disease status accurately. All CODs were checked twice by the study team, which was feasible for 300 CODs, but would have caused significant delays between participant recruitment and data collection if 600

needed to be collected and checked. By comparing the two samples, the hypothesis is tested that patient groups and preferences are similar, whether they self-report or not.

To account for potential scale heterogeneity across the physician's COD and no physician's COD samples and allow pooling, MXL models were estimated with fully correlated covariance matrices; this was done in a Bayesian framework for ease of computation. These two cohorts provided a comparison of preferences between self-reported and physician-confirmed patients, demonstrating that preferences were very similar, except for when it came to daily activities. This may be due to the fact that the physician's COD sample scored significantly higher migraine severity than the no physician's COD sample on every PRO instrument (MSSS, MSQ, MIDAS, PGI-S). Additionally, the physician's COD sample had more migraine headache days per month than the no physician's COD sample.

Additionally, hypothetical bias can occur in DCEs when patients respond to hypothetical scenarios in ways that do not reflect their actual responses in a clinical setting. However, the external validity of DCE methods has been shown to elicit trade-offs relevant to clinical decision-making⁵⁸ and aligned with real-world health choices.⁵⁹ To mitigate hypothetical bias by making the DCE credible and relevant to patients, the DCE design, question framing, and question wording were developed based on patient input from qualitative pilot interviews. Further, a tutorial was included in the warm-up to ensure that patients understood the attribute descriptions.

Conclusion

Patients with EM valued treatments that have a higher likelihood of reducing their migraine attacks. Other treatment attributes related to clinical efficacy were also important to patients, who highly valued a treatment's impact on daily activities and the time required for it to reach full effectiveness. While patients did not value a reduction in the need for acute medication as highly as other efficacy-related attributes, they still placed moderate value on this attribute. Patients were not concerned with the treatment's mode of administration. These results can be used to inform a patient-centered treatment strategy that considers the relative importance of treatment attributes in clinical decision-making. HCPs can use preference data to understand patient preferences for different attributes and ensure that treatment decisions align with their patients' individual preferences.

Abbreviations

CGRP, calcitonin gene-related peptide; CI, confidence interval; COD, confirmation of diagnosis; DCE, discrete choice experiment; EM, episodic migraine; FDA, US Food and Drug Administration; mAbs, monoclonal antibodies; MSQ, Migraine-Specific Quality of Life Questionnaire; MSSS, Migraine Symptom Severity Score; PGI-S, Patient Global Impression of Severity; PRO, patient-reported outcome; SBSQ, Set of Brief Screening Questions; MIDAS, Migraine Disability Assessment questionnaire; RAI, relative attribute importance; SD, standard deviation.

Data Sharing Statement

Lilly provides access to all individual participant data collected during the trial, after anonymization, with the exception of pharmacokinetic or genetic data. Data are available to request 6 months after the indication studied has been approved in the US and EU and after primary publication acceptance, whichever is later. No expiration date of data requests is currently set once data are made available. Access is provided after a proposal has been approved by an independent review committee identified for this purpose and after receipt of a signed data sharing agreement. Data and documents, including the study protocol, statistical analysis plan, clinical study report, blank or annotated case report forms, will be provided in a secure data sharing environment. For details on submitting a request, see the instructions provided at <u>www.vivli.org</u>.

Acknowledgment

Athanasia Benekou and Michael Franklin, of PPD a Thermo Fisher company, provided medical writing services, which were funded by Eli Lilly and Company, in accordance with Good Publication Practice (GPP) guidelines (Good Publication Practice [GPP] Guidelines for Company-Sponsored Biomedical Research: 2022 Update | Annals of Internal Medicine [acpjournals.org]).

Author Contributions

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

This study was funded by Eli Lilly and Company.

Disclosure

LV, OJV, and ATH are employees and/or stockholders of Eli Lilly and Company. CW, MQ, and MT are employees of Evidera Inc. which received payment from Eli Lilly for work relating to this study. The authors report no other conflicts of interest in this work.

References

- 1. Serrano D, Lipton RB, Scher AI, et al. Fluctuations in episodic and chronic migraine status over the course of 1 year: implications for diagnosis, treatment and clinical trial design. J Headache Pain. 2017;18(1):101. doi:10.1186/s10194-017-0787-1
- 2. The American Headache Society. Position statement on integrating new migraine treatments into clinical practice. *Headache*. 2019;59(1):1–18. doi:10.1111/head.13456
- 3. Seo J, Tervonen T, Ueda K, Zhang D, Danno D, Tockhorn-Heidenreich A. Discrete choice experiment to understand Japanese patients' and physicians' preferences for preventive treatments for migraine. *Neurol Ther.* 2023;12(2):651–668. doi:10.1007/s40120-023-00453-0
- 4. Hubig LT, Smith T, Chua GN, et al. A stated preference survey to explore patient preferences for novel preventive migraine treatments. *Headache*. 2022;62(9):1187–1197. doi:10.1111/head.14386
- 5. Mansfield C, Gebben DJ, Sutphin J, et al. Patient preferences for preventive migraine treatments: a discrete-choice experiment. *Headache*. 2019;59 (5):715–726. doi:10.1111/head.13498
- 6. Matza LS, Deger KA, Vo P, Maniyar F, Goadsby PJ. Health state utilities associated with attributes of migraine preventive treatments based on patient and general population preferences. *Qual Life Res.* 2019;28(9):2359–2372. doi:10.1007/s11136-019-02163-3
- 7. Schwedt TJ, Martin A, Kymes S, et al. Patient preferences for attributes of injected or infused preventive migraine medications: findings from a discrete choice experiment. *Headache*. 2023;63(4):484–493. doi:10.1111/head.14476
- Seo J, Smith CA, Thomas C, et al. Patient perspectives and experiences of preventive treatments and self-injectable devices for migraine: a focus group study. *Patient*. 2022;15(1):93–108. doi:10.1007/s40271-021-00525-z
- 9. Peres MF, Silberstein S, Moreira F, et al. Patients' preference for migraine preventive therapy. *Headache*. 2007;47(4):540–545. doi:10.1111/j.1526-4610.2007.00757.x
- 10. Cowan R, Cohen JM, Rosenman E, Iyer R. Physician and patient preferences for dosing options in migraine prevention. *J Headache Pain*. 2019;20 (1):50. doi:10.1186/s10194-019-0998-8
- 11. Ailani J, Winner P, Hartry A, et al. Patient preference for early onset of efficacy of preventive migraine treatments. *Headache*. 2022;62(3):374–382. doi:10.1111/head.14255
- 12. Blumenfeld AM, Bloudek LM, Becker WJ, et al. Patterns of use and reasons for discontinuation of prophylactic medications for episodic migraine and chronic migraine: results from the second international burden of migraine study (IBMS-II). *Headache*. 2013;53(4):644–655. doi:10.1111/ head.12055
- 13. Ray JC, Kapoor M, Stark RJ, et al. Calcitonin gene related peptide in migraine: current therapeutics, future implications and potential off-target effects. J Neurol Neurosurg Psychiatry. 2021;92(12):1325–1334. doi:10.1136/jnnp-2020-324674
- 14. Charles AC, Digre KB, Goadsby PJ, Robbins MS, Hershey A. The American headache society. calcitonin gene-related peptide-targeting therapies are a first-line option for the prevention of migraine: an American Headache Society position statement update. *Headache*. 2024;64(4):333–341. doi:10.1111/head.14692
- 15. Sacco S, Amin FM, Ashina M, et al. European Headache Federation guideline on the use of monoclonal antibodies targeting the calcitonin gene related peptide pathway for migraine prevention 2022 update. *J Headache Pain*. 2022;23(1):67. doi:10.1186/s10194-022-01431-x
- 16. Messina R, Huessler EM, Puledda F, Haghdoost F, Lebedeva ER, Diener HC. Safety and tolerability of monoclonal antibodies targeting the CGRP pathway and gepants in migraine prevention: a systematic review and network meta-analysis. *Cephalalgia*. 2023;43(3):3331024231152169. doi:10.1177/03331024231152169
- Haghdoost F, Puledda F, Garcia-Azorin D, Huessler EM, Messina R, Pozo-Rosich P. Evaluating the efficacy of CGRP mAbs and gepants for the preventive treatment of migraine: a systematic review and network meta-analysis of Phase 3 randomised controlled trials. *Cephalalgia*. 2023;43 (4):3331024231159366. doi:10.1177/03331024231159366
- 18. Schwedt TJ, Myers Oakes TM, Martinez JM, et al. Comparing the efficacy and safety of galcanezumab versus rimegepant for prevention of episodic migraine: results from a randomized, controlled clinical trial. *Neurol Ther.* 2024;13(1):85–105. doi:10.1007/s40120-023-00562-w
- 19. AIMOVIG (erenumab-aooe) injection for subcutaneous use. Initial U.S. approval: 2018 prescribing information. Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2021/761077s009lbl.pdf. Accessed February 27, 2024.
- 20. AJOVY (fremanezumab-vfrm) injection, for subcutaneous use Initial U.S. Approval: 2018. Prescribing Information. Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2018/761089s000lbl.pdf. Accessed February 27, 2024.

- EMGALITY (galcanezumab-gnlm) injection, for subcutaneous use. Initial U.S. Approval: 2018. Prescribing Information. Available from: https:// www.accessdata.fda.gov/drugsatfda_docs/label/2022/761063s006lbl.pdf. Accessed February 27, 2024.
- 22. VYEPTITM (eptinezumab-jjmr) injection for intravenous use. Initial U.S. Approval: 2020 Prescribing Information. Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2020/761119s000lbl.pdf. Accessed February 27, 2024.
- QULIPTA (atogepant) tablets, for oral use Initial U.S. Approval: 2021.Prescribing Information. Available from: https://www.accessdata.fda.gov/ drugsatfda_docs/label/2021/215206Orig1s000lbl.pdf. Accessed February 27, 2024.
- 24. NURTEC ODT. (rimegepant) orally disintegrating tablets, for sublingual or oral use. Initial U.S. Approval: 2020. Prescribing Information. Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2022/212728s009lbl.pdf. Accessed February 27, 2024.
- 25. European Medicines Agency (EMA). GPV Module VI Management and reporting of adverse reactions to medicinal products (Rev 1). Guideline on good pharmacovigilance practices. Available from: https://www.ema.europa.eu/en/documents/regulatory-procedural-guideline-good-pharmacovigilance-practices-gvp-module-vi-collection-management-submission-reports_en.pdf. Accessed March 28, 2019.
- 26. Official Journal of the European Union. Commission implementing regulation (EU) No 520/2012 of 19 June 2012 on the performance of pharmacovigilance activities provided for in Regulation (EC) No 726/2004 and Directive 2001/83/EC of the European Parliament and of the Council. Available from: https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2012:159:0005:0025:EN:PDF. Accessed March 28, 2019.
- 27. Bridges JF, Hauber AB, Marshall D, et al. Conjoint analysis applications in health--a checklist: a report of the ISPOR good research practices for conjoint analysis task force. *Value Health*. 2011;14(4):403–413. doi:10.1016/j.jval.2010.11.013
- 28. International Society for Pharmacoeconomics Outcomes Research (ISPOR). ISPOR good practices for outcomes research index. Available from: https://www.ispor.org/heor-resources/good-practices-for-outcomes-research. Accessed March 28, 2019.
- 29. Reed Johnson F, Lancsar E, Marshall D, et al. Constructing experimental designs for discrete-choice experiments: report of the ISPOR conjoint analysis experimental design good research practices task force. *Value Health*. 2013;16(1):3–13. doi:10.1016/j.jval.2012.08.2223
- 30. Chew LD, Bradley KA, Boyko EJ. Brief questions to identify patients with inadequate health literacy. Fam Med. 2004;36(8):588-594.
- Fransen MP, Van Schaik TM, Twickler TB, Essink-Bot ML. Applicability of internationally available health literacy measures in the Netherlands. J Health Commun. 2011;16(Suppl 3):134–149. doi:10.1080/10810730.2011.604383
- 32. Lipkus IM, Samsa G, Rimer BK. General performance on a numeracy scale among highly educated samples. *Med Decis Making*. 2001;21 (1):37-44. doi:10.1177/0272989X0102100105
- 33. Martin BC, Pathak DS, Sharfman MI, et al. Validity and reliability of the migraine-specific quality of life questionnaire (MSQ Version 2.1). *Headache*. 2000;40(3):204–215. doi:10.1046/j.1526-4610.2000.00030.x
- 34. Rendas-Baum R, Bloudek LM, Maglinte GA, Varon SF. The psychometric properties of the migraine-specific quality of life questionnaire version 2.1 (MSQ) in chronic migraine patients. *Qual Life Res.* 2013;22(5):1123–1133. doi:10.1007/s11136-012-0230-7
- 35. Speck R, Kudrow D, Christie S, Ayer D, Ford J, Bushnell D. The migraine-specific quality of life questionnaire, role function restrictive domain: defining clinically meaningful categories of functional impairment severity. J Headache Pain. 2021;22:P0226.
- 36. Guy W. ECDEU Assessment Manual for Psychopharmacology. US Department of Health, Education, and Welfare, Public Health Service ...; 1976.
- 37. Lipton RB, Stewart WF, Sawyer J, Edmeads JG. Clinical utility of an instrument assessing migraine disability: the Migraine Disability Assessment (MIDAS) questionnaire. *Headache*. 2001;41(9):854–861. doi:10.1111/j.1526-4610.2001.01156.x
- Serrano D, Buse D, Reed M, Runken M, Lipton R. Development of the Migraine Symptom Severity Score (MSSS): a latent variable model for migraine definition. PO-86. 52nd annual scientific meeting of the American-headache-society. June 24-27, 2010. Los Angeles, California, USA. *Headache*. 2010;V50(Sppl 1):S40.
- 39. Nguyen TC, Robinson J, Kaneko S, Nguyen TC. Examining ordering effects in discrete choice experiments: a case study in Vietnam. *Econ Anal Policy*. 2015;45:39–57. doi:10.1016/j.eap.2015.01.003
- 40. Croop R, Lipton RB, Kudrow D, et al. Oral rimegepant for preventive treatment of migraine: a Phase 2/3, randomised, double-blind, placebo-controlled trial. *Lancet*. 2021;397(10268):51–60. doi:10.1016/S0140-6736(20)32544-7
- 41. Ailani J, Lipton RB, Goadsby PJ, et al. Atogepant for the preventive treatment of migraine. N Engl J Med. 2021;385(8):695–706. doi:10.1056/ NEJMoa2035908
- 42. Skljarevski V, Matharu M, Millen BA, Ossipov MH, Kim BK, Yang JY. Efficacy and safety of galcanezumab for the prevention of episodic migraine: results of the EVOLVE-2 Phase 3 randomized controlled clinical trial. *Cephalalgia*. 2018;38(8):1442–1454. doi:10.1177/0333102418779543
- 43. Stauffer VL, Dodick DW, Zhang Q, Carter JN, Ailani J, Conley RR. Evaluation of galcanezumab for the prevention of episodic migraine: the EVOLVE-1 randomized clinical trial. JAMA Neurol. 2018;75(9):1080–1088. doi:10.1001/jamaneurol.2018.1212
- 44. Mulleners WM, Kim BK, Lainez MJA, et al. Safety and efficacy of galcanezumab in patients for whom previous migraine preventive medication from two to four categories had failed (CONQUER): a multicentre, randomised, double-blind, placebo-controlled, phase 3b trial. *Lancet Neurol*. 2020;19(10):814–825. doi:10.1016/S1474-4422(20)30279-9
- 45. Ford JH, Ayer DW, Zhang Q, et al. Two randomized migraine studies of galcanezumab: effects on patient functioning and disability. *Neurology*. 2019;93(5):e508–e517. doi:10.1212/WNL.00000000007856
- 46. ChoiceMetrics. Ngene 1.2 User Manual and Reference Guide. 2019.
- 47. Rose JM, Bliemer MC. Constructing efficient stated choice experimental designs. Transp Rev. 2009;29(5):587-617. doi:10.1080/01441640902827623
- 48. Clark MD, Determann D, Petrou S, Moro D, de Bekker-Grob EW. Discrete choice experiments in health economics: a review of the literature. *Pharmacoeconomics*. 2014;32(9):883–902. doi:10.1007/s40273-014-0170-x
- 49. Lancsar E, Louviere J. Deleting 'irrational' responses from discrete choice experiments: a case of investigating or imposing preferences? *Health Econ.* 2006;15(8):797–811. doi:10.1002/hec.1104
- 50. Reed Johnson F, Yang JC, Reed SD. The internal validity of discrete choice experiment data: a testing tool for quantitative assessments. *Value Health*. 2019;22(2):157–160. doi:10.1016/j.jval.2018.07.876
- 51. Thurstone LL. A law of comparative judgment. Psychol Rev. 1927;34(4):273-286. doi:10.1037/h0070288
- 52. McFadden D. Conditional Logit Analysis of Qualitative Choice Behaviour. Academic Press; 1974.
- 53. Manski CF. The structure of random utility models. Theory and Decision. 1977;8(3):229-254. doi:10.1007/BF00133443
- 54. Hess S, Train K. Correlation and scale in mixed logit models. J Choice Modelling. 2017;23:1-8. doi:10.1016/j.jocm.2017.03.001
- 55. Lancsar E, Fiebig DG, Hole AR. Discrete choice experiments: a guide to model specification, estimation and software. *Pharmacoeconomics*. 2017;35(7):697-716. doi:10.1007/s40273-017-0506-4

- 56. Burch R, Rizzoli P, Loder E. The prevalence and impact of migraine and severe headache in the United States: updated age, sex, and socioeconomic-specific estimates from government health surveys. *Headache*. 2021;61(1):60–68. doi:10.1111/head.14024
- 57. Stovner LJ, Nichols E, Steiner TJ, Global. regional, and national burden of migraine and tension-type headache, 1990-2016: a systematic analysis for the Global Burden of Disease Study 2016. *Lancet Neurol.* 2018;17(11):954–976. doi:10.1016/s1474-4422(18)30322-3
- de Bekker-Grob EW, Swait JD, Kassahun HT, et al. Are healthcare choices predictable? The impact of discrete choice experiment designs and models. *Value Health.* 2019;22(9):1050–1062. doi:10.1016/j.jval.2019.04.1924
- Quaife M, Terris-Prestholt F, Di Tanna GL, Vickerman P. How well do discrete choice experiments predict health choices? A systematic review and meta-analysis of external validity. *Eur J Health Econ.* 2018;19(8):1053–1066. doi:10.1007/s10198-018-0954-6

Patient Preference and Adherence

Dovepress Taylor & Francis Group

Publish your work in this journal

Patient Preference and Adherence is an international, peer-reviewed, open access journal that focusing on the growing importance of patient preference and adherence throughout the therapeutic continuum. Patient satisfaction, acceptability, quality of life, compliance, persistence and their role in developing new therapeutic modalities and compounds to optimize clinical outcomes for existing disease states are major areas of interest for the journal. This journal has been accepted for indexing on PubMed Central. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/patient-preference-and-adherence-journal

🖪 🗙 in 🗖